Author's response to reviews

Title: Endoscopic management of hypertensive intraventricular haemorrhage with obstructive hydrocephalus

Authors:

Yad Ram Yadav (yadavyr@yahoo.co.in)
Gaurav Mukerji (g.mukerji@imperial.ac.uk)
Ravikiran Shenoy (r.shenoy@imperial.ac.uk)
Abhijeet Basoor (drasbasoor@hotmail.com)
Gaurav Jain (gauraviain1000@yahoo.com)
Adam Nelson (drenelsonadam@yahoo.co.uk)

Version: 4 Date: 4 January 2007

Author's response to reviews: see over
Sub: Submission of revised manuscript

Dear Editor,

Please find the revised version of the manuscript with the formatting changes –

1. The Figure 3 of the previous submission is now named Figure 2 and the Figure 2 is now Figure 3 to ensure that the figures are referenced in numerically consecutive order. Corresponding changes have been made in the legends and their reference in the text of the manuscript.

2. Upper case of the ‘TABLE’ has been changed to ‘Table’.

3. ‘None of the authors or the study was externally funded’ – has been declared in the acknowledgement section.

We hope that this manuscript now conforms to the Journal’s format requirements and is acceptable for publication.

Yours sincerely,

GAURAV MUKERJI